Author's response to reviews

**Title:** A European study on alcohol and drug use among young drivers: the TEND by Night study design and methodology.

**Authors:**

Roberta Siliquini (roberta.siliquini@unito.it)
Simone Chiadò Piat (simone.chiado@unito.it)
Francisco Alonso (francisco.alonso@uv.es)
Axel Druart (axel@ryd.nl)
Marcin Kedzia (mkedzia@holowczyc.pl)
Antonio Mollica (guidasicura@motoroasi.com)
Valeria Siliquini (v.siliquini@setinweb.it)
Daniel Vankov (chairman@open-youth.org)
Anita Villerusa (anita.villerusa@rsu.lv)
Lamberto Manzoli (lmanzoli@post.harvard.edu)
TEND Group "TDG" (info@tendbynight.eu)

**Version:** 4 **Date:** 13 April 2010

**Author's response to reviews:** see over
Dear Editor,

I'm happy to submit our revised methodological manuscript “A European study on alcohol and drug use among young drivers: the TEND by Night study design and methodology” for the publication on "BMC Public Health”, after we made the final formatting changes.

In particular the changes made were:

- the name of the study group has been included within the list of authors in the online revision process;
- the names of the members of the study group have been removed from the title page and placed in the Acknowledgements section;
- the correct functioning link to website has been provided in Reference 36;
- the full citation details for Reference 40 have been provided;
- the manuscript has been checked to make any final correction.

Thank you very much for your attention.

Looking forward to hearing from you,

Yours sincerely

Simone Chiadò Piat